Author’s response to reviews

Title: Health promoting lifestyle of university students in Saudi Arabia: a cross-sectional assessment

Authors:

Khalid Imutairi (Kalmutairim@ksu.edu.sa)
Wadi Alonazi (waalonazi@KSU.EDU.SA)
Jason Vinluan (jvinluan@ksu.edu.sa)
Turky Almigbal (almogbal@yahoo.com)
Mohammed Ali Batais (drmohammed34@gmail.com)
Abdulaziz Alodhayani (drodhayani12@yahoo.com)
Norah Alsadhan (nalsadhan@ksu.edu.sa)
Regie Tumala (rtumala@KSU.EDU.SA)
Mahaman Moussa (mmoussa@KSU.EDU.SA)
Ahmad Aboshaiqah (aaboshaiqah@ksu.edu.sa)
Razan Ibrahim Alhoqail (razan724@gmail.com)

Version: 3 Date: 23 Aug 2018

Author’s response to reviews:

1. Declarations and heading

Please provide all sections listed under the heading Declarations in our Submission guidelines: https://bmcoralhealth.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article

Please note also that any Declarations sections which are not applicable to your manuscript should still be included with the statement "Not applicable".

Please ensure your Declarations section includes the following subsections:

- Ethics approval and consent to participate
- Consent to publish
- Availability of data and materials
- Competing interests
- Funding
- Authors' Contributions
- Acknowledgements
- Authors' Information

2. Ethics approval and consent to participate

Please include an Ethics Approval and Consent to Participate declaration section in your manuscript, in which you confirm that your study was submitted to and approved by your institutional ethics committee (named), as indicated in your Methods section. Please also add a statement to confirm that informed consent, written or verbal, was obtained from all participants and clearly state this in your manuscript. If verbal, please state the reason and whether the ethics committee approved this procedure. If the need for consent was waived by an Institutional Review Board (IRB) or is deemed unnecessary according to national regulations, please clearly state this, including the name of the IRB or a reference to the relevant legislation.

- Thank you for your comment. We already add the Ethics approval and consent to participate in the manuscript.

3. Funding

We note that you have not included a ‘Funding’ section in the Declarations. All sources of funding for the research reported should be declared. If no funding was obtained for your study we still require this section to be included with the statement “No funding was obtained for this study”.

- The source of funding was added in the Funding section.

4. Authors' contributions

Please include a declarations section for authors' contributions, in which you individually name (using initials, rather than full names) each author and include their contributions to this manuscript, following the guidelines below. Please also include a statement in the Authors'
contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

- Each author's contribution was included in the manuscript following the journal guidelines.

Guidelines:

An 'author' is generally considered to be someone who has made substantive intellectual contributions to a published study. According to the ICMJE guidelines, to qualify as an author one should have:

a) made substantial contributions to conception and design, or acquisition of data, or analysis and interpretation of data; AND

b) been involved in drafting the manuscript or revising it critically for important intellectual content; AND

c) given final approval of the version to be published. Each author should have participated sufficiently in the work to take public responsibility for appropriate portions of the content; AND

d) agreed to be accountable for all aspects of the work in ensuring that questions related to the accuracy or integrity of any part of the work are appropriately investigated and resolved.

Anyone listed as an author must be included in this section. If you choose to change your author list you will need to fill out a change in authorship form and send it by email to the Editorial office to be approved by the Editor. The form can be found here: https://www.biomedcentral.com/getpublished/editorial-policies#authorship.

Anyone who contributed towards the article who does not meet the criteria for authorship can be acknowledged in the ‘Acknowledgements’ section.

5. Acknowledgements

Please remove the funding information from the Acknowledgements and include it in the Funding section instead. If you have no further acknowledgements please put “Not Applicable” in the Acknowledgements section.

- We already remove the funding information from the Acknowledgement section and include it in the Funding section.

Thank you